Author's response to reviews

Title: Clinical features and dysfunctions of iron metabolism in Parkinson disease patients with hyper echogenicity in substantia nigra: a cross-sectional study

Authors:
Shu-yang Yu (yushuyang615@163.com)
Chen-Jie Cao (iceblue0703@yahoo.cn)
Li-jun Zuo (zuolijun1987@126.com)
Ze-jie Chen (chenzejie55@sina.com)
Teng-hong Lian (lth1990@126.com)
Fang Wang (wftw123@163.com)
Yang Hu (1132471469@qq.com)
Ying-shan Piao (piaoyingshan@aliyun.com)
Li-xia Li (lixia188@163.com)
Peng Guo (pengpeng5529@126.com)
Li Liu (qingyangxiaoli@163.com)
Qiu-jin Yu (qiujin2014@163.com)
Rui-dan Wang (wrd-1213@163.com)
Piu Chan (pbchan90@gmail.com)
Sheng-di Chen (chen_sd@medmail.com.cn)
Xiao-min Wang (xmwang@ccmu.edu.cn)
Wei ZHANG (ttyyzw@163.com)

Version: 2 Date: 17 Oct 2017

Author's response to reviews:
Dear editor in-chief of BMC Neurology:
Please find the enclosed manuscript entitled “Clinical features and dysfunctions of iron metabolism in Parkinson disease patients with hyper echogenicity in substantia nigra: a cross-sectional study”, by Shu-Yang YU et al., to be submitted as an original article to BMC Neurology for the consideration of publication. The following are the answers for the questions raised by the reviewers.

Reviewer reports:

(Reviewer 1): The authors have answered most of the queries. However, there is no illustration of grade IV in figure 1 and the figure legend of figure 1 is missing.

We have revised Figure 1, including all illustrations of Grade I to Grade V, and also added figure legend, see Figure 1.

(Reviewer 2): The authors replied all my questions well and I had no more questions.

Sincerely yours,

Wei ZHANG, on behalf of the authors